

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Almerud, Jakob; Krygier, Dominika; Lundvall, Henrik; Njie, Mambuna

Working Paper Measuring Riksbank monetary policy: Shocks and macroeconomic transmission

Sveriges Riksbank Working Paper Series, No. 445

Provided in Cooperation with: Central Bank of Sweden, Stockholm

Suggested Citation: Almerud, Jakob; Krygier, Dominika; Lundvall, Henrik; Njie, Mambuna (2024) : Measuring Riksbank monetary policy: Shocks and macroeconomic transmission, Sveriges Riksbank Working Paper Series, No. 445, Sveriges Riksbank, Stockholm

This Version is available at: https://hdl.handle.net/10419/310405

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Measuring Riksbank Monetary Policy: Shocks and Macroeconomic Transmission

Jakob Almerud, Dominika Krygier, Henrik Lundvall and Mambuna Njie December 2024

WORKING PAPERS ARE OBTAINABLE FROM

www.riksbank.se/en/research

Sveriges Riksbank • SE-103 37 Stockholm Fax international: +46 8 21 05 31 Telephone international: +46 8 787 00 00

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The opinions expressed in this article are the sole responsibility of the author(s) and should not be interpreted as reflecting the views of Sveriges Riksbank.

Measuring Riksbank Monetary Policy: Shocks and Macroeconomic Transmission

Jakob Almerud⁺

Dominika Krygier[†]

Henrik Lundvall⁺

Mambuna Njie[‡]

Sveriges Riksbank Working Paper Series

No. 445

December 2024

Abstract. We construct and make available a new monetary policy event study database with high-frequency financial market reactions to Riksbank communications, spanning a period of 20 years. Using these data as instruments, we estimate the macroeconomic effects of monetary policy shocks in Sweden. A temporary, unexpected policy rate tightening induces an immediate and persistent appreciation of the krona exchange rate, as well as a gradual, negative response in output and consumer prices. These results are statistically significant, economically meaningful and robust to a number of variations in our econometric specification. In particular, we consider the possibility that financial market reactions to Riksbank communications may consist not only of pure monetary policy shocks, but could also reflect market participants' updates concerning the central bank's reaction function.

Keywords: monetary policy surprise database, monetary policy shocks, intraday, event study, proxy VAR, macroeconomic effects

JEL classification: E43, E44, E52, E58, G14

⁺ Sveriges Riksbank, Monetary Policy Department, <u>Jakob.Almerud@riksbank.se</u>, <u>Dominika.Krygier@riksbank.se</u>, <u>Henrik.Lundvall@riksbank.se</u>, [‡] At the time of writing, Mambuna Njie worked at the Monetary Policy Department of the Riksbank. The authors want to thank Jens Iversen, Mathias Klein, Stefan Laséen, Ulf Söderström, Ingvar Strid, Anders Vredin and seminar participants at Sveriges Riksbank for helpful comments. The opinions expressed in this article are the sole responsibility of the authors and should not be interpreted as reflecting the views of Sveriges Riksbank.

1 Introduction

What are the effects of monetary policy on key macroeconomic variables such as output, inflation and unemployment? This is a longstanding and frequently revisited question in the monetary economics literature. Following important contributions in the early 2000s, e.g. by Kuttner (2001) and Faust et al. (2004), it has become standard to use high-frequency (HF) data from financial markets to identify the effects of monetary policy, since such instruments-based analysis can convincingly say something about the causal effects of changes in the stance of monetary policy. While an extensive literature has used HF instruments to study the effects of U.S. monetary policy, comparatively less research has been devoted to other monetary jurisdictions, partly because of data availability issues. In recent years, however, researchers have constructed novel monetary policy event study databases in order to facilitate a systematic analysis of monetary policy in the euro area (EA) and the United Kingdom (UK).²

This paper contributes to this literature in several ways. First, we construct and make available a Riksbank monetary policy event study database, featuring financial market reactions during short time windows from days when the Riksbank, Sweden's central bank, announced monetary policy decisions or released minutes from monetary policy meetings. We include reactions in seven different interest rate instruments, two exchange rates and an index of the Nasdaq Stockholm stock exchange. The database spans a period of 20 years, from 2004 to 2024, and we expect it to be continually updated going forward, thereby facilitating and stimulating research into the effects of monetary policy in Sweden.

Second, we use the new database to identify monetary policy shocks and estimate how these impact key macroeconomic variables. Following Swanson (2021) and Gürkaynak et al. (2005b), we apply factor analysis to the set of seven interest rate swaps of different maturities to identify and distinguish different types of shocks: shocks to the current level of the policy rate, shocks to its expected future level and shocks related to the Riksbank's use of quantitative monetary policy measures. We focus on the first two of these types of unexpected changes in Riksbank monetary policy and use them as external instruments in a structural VAR-model.

Concerning shocks to the current level of the policy interest rate (factor 1), the results show that an unexpected, temporary tightening (loosening) of the policy rate induces an immediate and persistent appreciation (depreciation) of the krona exchange rate, as well as a more gradual, negative (positive) effect on output and consumer prices. These effects are statistically

² See Altavilla et al. (2019) and Braun et al. (2024).

significant, economically meaningful and robust to a number of variations in our econometric specification.

According to our baseline estimate, shocks to the future expected level of the policy interest rate (factor 2) imply a gradual increase (decrease) of the policy interest rate over a period of about 18 months, followed by a gradual decrease (increase). The effects on output and consumer prices are similar to those of an unexpected, temporary change in the current level of the policy rate (factor 1). In the case of shocks to the expected future level of the policy rate (factor 2), the instrument is however on the weak side, and the results are less robust.

Besides usual robustness exercises, e.g. such related to the specification of the VAR-model, we take special care to investigate robustness to alternative interpretations of the financial market reactions embedded in the new event study database. Several studies document that financial market reactions – or surprises – to Federal Reserve announcements are autocorrelated, and also correlated with macro-financial information that was publicly available to market participants prior to the announcements (e.g. Cieslak (2018), Miranda-Agrippino and Ricco (2021), Bauer and Swanson (2023)). These findings call into question the exogeneity of these surprises, and thus the validity of the practice of using them as instruments for monetary policy shocks. The literature has proposed different interpretations of these correlations. Two prominent such interpretations are the "Fed information effect"-hypothesis, as in Campbell et al. (2012), Nakamura and Steinsson (2018) and Miranda-Agrippino and Ricco (2021), and the "Fed response to news channel", proposed by Bauer and Swanson (2023a and 2023b).³

The Fed information effect-hypothesis starts from the assumption that the central bank may have private information about the state of the economy, and that this information may be disseminated to market participants through announcements of monetary policy decisions. If such an information channel exists, monetary policy surprises may derive not only from true monetary policy shocks, but could also be the result of news about such private information. Laséen (2020) investigates if any such "Riksbank information effect" is relevant in the case of Swedish monetary policy surprises. Using Riksbank forecasts to construct an informationallyrobust HF instrument for Swedish monetary policy shocks, Laséen (2020) finds that central bank information shocks do not materially confound VAR-estimates of the macroeconomic transmission of Swedish monetary policy shocks, and suggests that the lack of a clear

³ The idea that the central bank may have private information about the state of the economy, and that such information may be transmitted to market participants through central bank announcements, was suggested by Romer and Romer (2000). Ellingsen and Söderström (2001) investigated the same idea, both theoretically and empirically, with a focus on the response of long-maturity interest rates to the release of central bank information.

information channel may be related to the Riksbank's transparency and communication strategies.

In this study, we instead focus on another potential source of bias when financial market surprises are used as instruments for monetary policy shocks, namely the "Fed response to news channel", proposed by Bauer and Swanson (2023a and 2023b). The basic idea is that market participants only have imperfect knowledge of the true parameters of the central bank's reaction function. Using a simple theoretical model of learning, Bauer and Swanson (2023b) show that if those parameters change over time, the learning process can be a source of systematic surprises. In the model, market participants continually use central bank communications to update their estimates of the parameters of the central bank's monetary policy rule, thus creating *ex post* correlation between surprises and information that was publicly available at the time of those communications.

We investigate the relevance of such a "Riksbank response to news" channel for Sweden, thereby complementing the work of Laséen (2020). In order to distinguish true Riksbank monetary policy shocks from possible surprises due to updating, we construct a dataset of real-time macro-financial data that was available to market participants at each individual communication event that we use in our estimations. We find evidence of statistically significant correlation between the surprises and such real-time data, most notably with the krona exchange rate and with foreign rates of inflation. As a robustness exercise, and following Bauer and Swanson (2023b), we use as our instrument only that component of surprises that is orthogonal to these correlated variables. We find that cleaning our monetary policy surprises in such a way does not alter the conclusions about the macroeconomic effects of Riksbank monetary policy in any substantial way.

Our paper relates to other studies that use a HF approach to estimate the transmission of Swedish monetary policy shocks. As previously mentioned, Laséen (2020) studies the effects of monetary policy on economic activity and asset prices in Sweden, separately identifying the effects of a surprise change in the policy rate from effects of new information about economic fundamentals. Other studies that focus on macroeconomic responses are Brubakk et al. (2022), Amberg et al. (2022), Kilman (2022) and Laséen and Nilsson (2024). Other studies focus on the effect of monetary policy on financial variables: Fransson and Tysklind (2016), Iversen and Tysklind (2017), Åhl (2017), De Rezende (2017), Sandström (2018), Natvik et al. (2020) and Brubakk et al. (2021). Similar dynamics in other small open economies are examined by, for example, Braun et al. (2023), Cesa-Bianchi et al. (2020), Hambur and Haque (2023) and Soosalu (2024).

The rest of the paper is organised as follows. Section 2 contains a description of the Riksbank monetary policy event study database, including information about the monetary policy decision and communication cycle, and how monetary policy surprises are constructed. Section 3 describes how we use the database to identify monetary policy shocks in Sweden. Section 4 presents the effects of monetary policy shocks on macroeconomic variables, including the results of a few robustness exercises. Finally, section 5 concludes.

2 The Riksbank monetary policy event study database

This section describes the data and procedure used to extract financial market surprises (HF surprises) on days when the Riksbank released communication about monetary policy. It also includes a short description of the Riksbank's schedule for deciding and communicating Swedish monetary policy. The first vintage of the database, made available together with this paper, spans a period of 20 years, from August 20, 2004 through August 26, 2024.⁴ It includes surprises in interest rates from 269 communication events, including announcements from 121 regular monetary policy meetings, 24 extraordinary meetings, as well as the release of minutes from 124 meetings.⁵ In the database, we also include reactions of the SEK/USD and the SEK/EUR exchange rates, as well as the OMX Stockholm all-share index, which includes all the shares listed on the Nasdaq OMX Nordic Exchange Stockholm.

Throughout the period covered by the database, Swedish monetary policy was characterized by an official, announced 2 per cent inflation target and a flexible exchange rate regime. The Riksbank announced the inflation target in 1993, following the abrupt abandonment in 1992 of the previous fixed exchange rate regime.⁶ Credibility for the new regime increased gradually in the following years and was formally codified in 1999 with the introduction of new legislation, granting the central bank a high degree of institutional independence.⁷

⁴ The database on Riksbank monetary policy surprises can be found here: <u>https://www.riksbank.se/en-gb/about-the-riksbank/the-tasks-of-the-riksbank/research/database-for-event-studies-of-swedish-monetary-policy/</u>

⁵ On two of the 269 dates, 2024-03-13 and 2020-03-26, an announcement from an extraordinary meeting was released on the same day as the release of minutes. We treat those two dates as two communication events.

⁶ In January 1993, the Riksbank announced that a 2 per cent inflation target would apply, starting from 1995.

⁷ The Riksbank Act from 1999 stipulated a single mandate of price stability for Swedish monetary policy. However, wordings included in the preparatory work of the Act were interpreted by the Riksbank as a mandate for flexible inflation targeting. A new Riksbank Act came into force in January of 2023. The new legislation provides a non-ambiguous support for a regime of flexible inflation targeting. The 2023 Act stipulates that the overriding objective of the Swedish central bank is to maintain permanently low and stable inflation, but the Act also states: "Without neglecting the price stability objective, the Riksbank shall also contribute to a balanced development of production and employment (consideration for the real economy)". For further details, see Sveriges Riksbank (2023a).

2.1 The Riksbank monetary policy decision and communication cycle

While the number of pre-announced, regular monetary policy meetings has changed somewhat over the years, the schedule for the communication of the monetary policy decisions has remained fairly stable. From 2004 to 2006, seven to eight regular, pre-announced monetary policy meetings where scheduled per year, with a report being prepared by staff for every second meeting. Since 2007, a monetary policy report (MPR) or a shorter monetary policy update (MPU) has been prepared for every regular meeting, while the number of such regular meetings has varied between five and eight. The current practice, introduced this year, involves eight regular meetings, with four MPRs and four MPUs.

In the morning, the day after a regular meeting, normally at 09:30 a.m., the decision is announced in a press release that contains a brief motivation (one to two pages). The press release also contains information about any formal reservations entered against the decisions, if such exist.⁸ The MPR or MPU is released at the same time as the press release, giving the public immediate access to in-depth information about the economic outlook, risk assessments and other information that was relevant for the decision. Later in the day, usually at 11:00 a.m., a press conference is held with the governor and the head of the monetary policy department. One to two weeks after the meeting, minutes from the meeting are released, usually also at 9:30 a.m.⁹

In addition to reactions to the announcement of 121 regular monetary policy meetings, the database also contains announcements of 24 extraordinary meetings, most of which were held during the financial crisis of 2008-09 and the coronavirus pandemic (2020-21). The communication of decisions taken at those extraordinary meetings did not, in general, follow the communication schedule used for regular meetings. Instead, those decisions were announced on more of an *ad hoc* basis.

Previous studies that use HF surprises from Riksbank announcements have focused only on days when monetary policy decisions were communicated. Because we want to study the multidimensional nature of the Riksbank's monetary policy communication, there is a need to increase the number of observations relative to those of previous studies. Therefore, we also

⁸ For all but five of the regular meetings in the database, the decision was announced at 09:30 a.m. on the day after the meeting was held. On one occasion in 2004 (October meeting), two occasions in 2005 (March and October meetings), as well as two occasions in 2006 (February and October meetings), the announcement happened at an earlier time. The reason was that on those occasions, the governor was scheduled to appear at committee hearings in the Riksdag, the Swedish Parliament, the same morning.

⁹ The time lag between regular meetings and the publication of minutes has also varied over the years. The current practice involves a lag of one week.

include HF surprises from days when minutes were released. The minutes normally contain a fairly detailed account of the motivations that lead to the decisions that were taken, as well as information about the differences in views that may exist within the executive board. This type of information is of obvious relevance for financial market participants and analysts. Other types of communication from the Riksbank, e.g. the press conferences and speeches given by executive board members, are likely to contain additional information that is relevant for financial markets. We leave it to future research to further enrich the database and analysis in this dimension.

2.2 Data and the construction of surprises

We construct our HF monetary policy surprise database by using intraday tick data for seven different interest swap instruments reflecting expected interest rates at different horizons: the 1-month, 3-month and 6-month SEK overnight index swap (STINA) rates, as well as the 1-year, 2-year, 5-year and 10-year swap rates.^{10,11} This data comes from Datascope Tick History, provided by the London Stock Exchange Group (LSEG). STINA swap rates, which settle on the *Tomorrow/Next Stockholm interbank offered rate* (STIBOR), reflect short-term interest rate expectations among SEK money market participants. However, the STINA market was generally not very liquid in the early 2000s and it is therefore not meaningful to include Riksbank announcements before August 2004. Even within our 2004-2024 sample, there are announcement days and STINA contracts for which only a limited number of quotes exists, particularly in the early years of the sample. The longer maturity swap rates (1-10 year) settle on the 3-month STIBOR and the market for these contracts is considerably more liquid.

We follow a simple event study approach where we observe how the seven different swap rates adjust around a narrow window of time in each of our communication events. Specifically, we have two event windows around each communication event: one precommunication window that starts 10 minutes before the actual announcement or release, and

¹⁰ To be more precise, swap rates may reflect both an interest rate expectation and a (possibly time varying) premium. When constructing surprises, we compute the difference between the market rates prevailing in a tight window after and before the relevant Riksbank announcement. Following Gertler and Karadi (2014) and others, we assume this premium does not change materially in such a short window of time, and we may thus think of these differences in swap rates as measures of market surprises in interest rate expectations. The 1-month STINA contract (and the calculated surprises in this instrument on the day of a regular monetary policy announcement) reflects an average over the next month, meaning that part of the period is influenced by the old policy rate and part by the new rate just announced. To isolate the portion of the surprise that reflects the market's reaction to the new announced policy rate, the surprise is scaled using an adjustment factor. The adjustment factor represents the proportion of time (out of the total contract length, i.e. 30 days) during which the new announced policy rate applies, i.e. *adjustment factor = (30-(days between announcement and implementation))/30.*

¹¹ An interest rate swap is a financial instrument where parties exchange interest flows with each other. Most commonly, the parties exchange a fixed rate with a floating rate, which is determined by a reference rate. In Swedish kronor, there are OIS contracts against two reference rates, STIBOR T/N and SWESTR. The swaps against STIBOR T/N are usually referred to as STINA swaps. Other than OIS contracts, there are interest rate swaps against other reference rates. In Sweden, most interest rate swaps (which are not OIS contracts) are against STIBOR 3m.

ends at the actual announcement, which normally takes place at 9:30 a.m.; and one post-event window starting at the time of the announcement and ending 20 minutes later. We pick the median price within each window and then exploit the discontinuity in the (median of the) quoted prices of our seven financial instruments, calculating the difference between the price in the post-announcement window and the pre-announcement window.¹² Figure 1 illustrates the measurement windows used when calculating the monetary policy surprises.



Figure 1: Measurement windows for monetary policy surprises

Note. The figure sketches the measurement windows used for calculating the monetary policy surprises. We use two windows a pre-announcement window starting 10 minutes before the announcement, and a post-announcement window, starting 20 minutes after the announcement. The median price (average of the bid and ask prices) for each of the instruments in each window is used as the pre and post announcement price.

To illustrate the importance of using intraday data, and to get a sense of how the raw data looks like, Figure 2 shows the changes in two of our instruments on November 23, 2023. On that day, at 09:30 a.m. the Riksbank announced the decision from its last monetary policy meeting of that year, which had taken place the day before, on November 22. Prior to the announcement, most analysts expected the Riksbank to hold the policy rate unchanged at 4.00 per cent, after a year and a half of successive rate hikes. However, some Swedish economists argued that, given the high inflation and the relatively weak exchange rate, a further increase to 4.25 per cent would be appropriate.

¹² As already mentioned previously, the seven swap instruments are not equally liquid. For the long term interest rates, like the 1-10 year swap rates, there is a sufficient number of quotes in our data to enable us to look at short event windows of 10 and 20 minutes. For the shortest swap contracts, however, we have to adapt by expanding the window somewhat in order to capture any movement in prices. In some cases, the daily close and opening prices are used instead to calculate the change. In the cases where the window needs to be expanded, our criterion is that there has to be at least three quotes in each window to calculate a value for that window (a median quote will be calculated). If there are less than three quotes the window will expand until the criterion of three quotes is met.



Figure 2: Intraday movements of interest rate swaps on November 23, 2023

Note: The figure shows intraday quotes for two SEK and EUR interest rate swaps on November 23, 2023 when the Riksbank announced its decision to hold its policy rate at 4 per cent. Each dot is a quote from a market maker and the vertical line shows the timestamp of the announcement (9:30 am). Source: LSEG Datascope Tick History.

Before the announcement on November 23, market prices also indicated a high probability of the policy rate being kept unchanged at 4.00 per cent. This expectation is reflected in the price of the 1-month OIS contract. The quotes from the morning of November 23, which were slightly above 4.00 per cent, indicated that market participants saw some probability of an additional 25 basis point hike to 4.25 per cent. At 09:30, the Riksbank announced that the policy rate would be kept unchanged at 4.00 per cent. The accompanying monetary policy report made only small adjustments to the Riksbank's policy rate forecast. As shown in both figures, both the 1-month OIS and the 2-year swap rates fell immediately after the announcement, indicating that market participants perceived the Riksbank's communication as being slightly dovish.

Figure 2 also shows, in the two lower panes, how comparable interest rates in the euro area moved during the same day. Not surprisingly, there was no movement at all in the 1-month Euro OIS rate.¹³ However, the same morning, several European purchasing managers' indices

¹³ The 1-month OIS contract is closely tied to expectations of how the ECB deposit rate will evolve in the coming 30 days, and since there was no relevant news on that day that shifted market expectations concerning the ECBs December rate decision, the OIS-rate stayed completely flat.

(PMI) were released, with a surprisingly strong reading of the euro area PMI becoming public at 10 a.m. Shortly thereafter, medium- and long-term euro area and Swedish rates started to drift upwards. The parallel movements of the euro area and Swedish 2-year swap rates, as shown in Figure 2, highlights the strong connection between Swedish medium- to long-term rates and those of the euro area, Sweden's major trading partner. The figure illustrates the potential importance of using intraday data for measuring financial market surprises in Sweden, especially for durations beyond 6 to 12 months. The top-right pane in Figure 2 clearly shows how the market reacted to the 09:30 a.m. Riksbank announcement – an effect that would have been missed if daily data were used instead.

3 Using the database to identify monetary policy shocks in Sweden

3.1 Identifying shocks to monetary policy

For U.S. monetary policy, Gürkaynak et al. (2005b) and Swanson (2021) argue that monetary policy decisions and communications of the Federal Reserve affect asset prices in a multidimensional way. For example, some decisions and communications may have a relatively large effect on short-run interest rate expectations and only small effects on longer-run expectations; other decisions and communications may have a relatively large effect only on longer-run expectations, etc. Gürkaynak et al. (2005b) and Swanson (2021) use a static factor model to analyse the systematic part of these effects, and we apply the same method on our dataset of Swedish monetary policy surprises. The aim is to identify different types of monetary policy shocks.¹⁴ We estimate the following model

$X = F\Lambda + \varepsilon$

where *X* is a *T* × *N* matrix containing the calculated surprises of N=7 interest rate instruments during *T* = 265 monetary policy communication events in our sample (see section 2.2 for a description of the instruments), *F* is a *T* × *m* matrix of unobserved factors ($m \le N$), Λ is a $m \times N$ matrix of unobserved factor loadings and ε is a *T* × *N* matrix of white noise disturbances.

We then apply the Cragg and Donald (1997) matrix rank test and find evidence for the presence of three distinct factors in our dataset of Swedish monetary policy surprises.¹⁵ These factors remain abstract until we impose an economic interpretation, based on the types of shocks we believe may arise. The aim is therefore to structurally identify the factors without changing the original statistical decomposition. We do this by finding an orthogonal matrix *U* and rotated factors \tilde{F} such that $X = FUU^{-1}\Lambda + \varepsilon$.¹⁶

¹⁴ Before using the surprises from the database in our analysis, we remove four dates: 2008-10-08, 2008-10-15, 2020-09-15 and 2021-03-10. The announcement on October 8, 2008, was part of a co-ordinated easing of monetary policy by several central banks in the world, including the Riksbank. The minutes from this meeting were released a week later on October 15. On September 15, 2020 and March 10, 2021, the Riksbank decided to continue to offer its monetary policy counterparties loans in US dollars against collateral up to March 31, 2021. On March 10, 2021, this decision was extended until September 2021. Both these dates are excluded from our analysis due to the fact that this agreement was part of a swap arrangement between the Federal Reserve and a number of other central banks around the world, aimed at improving liquidity conditions in global financial markets, and hence not a decision aimed to affect monetary policy in Sweden per se.

¹⁵ See Appendix B for details of the test.

¹⁶ We tested for autocorrelation in the factors using the Durbin-Watson, Breusch-Godfrey, and Ljung-Box tests (up to 10 lags). In all cases, we failed to reject the null hypothesis of no autocorrelation, indicating that the data does not exhibit significant evidence of autocorrelation.

By defining $\tilde{F} = FU$ and $\hat{\Lambda} = U^{-1}\Lambda$, we identify the rotated factors and loadings. We impose the following restrictions on *U* to distinguish between different, orthogonal types of monetary policy shocks:

- The second factor is restricted to only influence future rate expectations, i.e. it cannot load on the 1-month OIS rate. We name this factor the **path factor**.¹⁷
- 2. The third factor is also not allowed to load on the 1-month OIS rate. Furthermore, the variance of the third factor is minimized during the periods 2004-08 to 2008-09, 2010-10 to 2014-04 and 2017-04 to 2020-02. We name this factor **quantitative measures factor**.
- 3. The first factor is associated with immediate policy rate changes. We therefore name this factor the **target factor**.¹⁸

The exact formulation of the second restriction may be viewed as an adaptation to Swedish conditions of the restrictions applied in Swanson (2021). During the period covered by our sample, the Riksbank has repeatedly used its balance sheet for monetary policy purposes, e.g. by extending special loans to its monetary policy counterparts and by conducting QE and QT operations. We therefore choose to interpret the third factor in a somewhat different way, compared to Swanson (2021), and accordingly label it 'quantitative measures factor'. Table 1 presents the factor loadings of the target factor, the path factor and the quantitative measures factor that result from the above identifying assumptions.

	111010	21 20111111					
	1-month OIS	3-month OIS	6-month OIS	1-year swap	2-year swap	5-year swap	10-year swap
Target factor	4.87	4.00	2.03	2.24	2.05	1.63	0.97
Path factor	0.00	0.26	1.01	1.67	2.03	1.86	1.43
Quantitative measures factor	0.00	1.34	2.59	1.02	0.76	0.22	0.00

Table 1. Estimated factor loadings

Note. Basis points. The table shows the factor loadings for our sample of 265 Riksbank announcements used in our analysis. The factor loadings are obtained by regressing the surprises onto the factors. The factor loading represents the effect, in basis points, of a one standard deviation shock to the factor. The 1-, 3- and 6-month OIS rates settle on STIBOR T/N, the 1, 2, 5 and 10-year swap rates settle on STIBOR 3m.

Based on the estimated factor loadings in Table 1, the target factor exhibits the largest effect at the short end of the yield curve, particularly on the 1-month OIS rate with a loading of 4.87. This means that a one standard deviation increase in the target rate factor is estimated to increase the current 1-month OIS rate by almost five basis points. This effect gradually diminishes along the curve, reducing to 0.97 for the 10-year swap rate. This pattern aligns with

¹⁷ The assumption that the second factor does not load on the 1-month OIS rate is due to the fact that this rate is priced as the compounded average overnight rate over 30 days. Since the Riksbank almost never, possibly except for extraordinary meetings, has two monetary policy meetings within a 30 day period, the 1-month OIS rate therefore cannot be impacted by what the Riksbank expects to do in the near-medium term (i.e. forward guidance).

¹⁸ No restrictions are placed on this factor (factor 1).

expectations, as changes in the current level of the policy rate primarily impact short-term interest rates before their influence tapers off over longer maturities. The results are also broadly consistent with the estimates in Gürkaynak et al. (2005b) and Swanson (2021). Shocks to the path factor have no effect on the 1-month OIS rate, by construction. Such shocks primarily affect the longer end of the yield curve, peaking around the 2-year horizon with a loading of 2.03. Also, these results are in line with previous studies. The quantitative measures factor yields somewhat different results compared to the first two factors. Like the path factor, the effect of a shock to the quantitative measures factor is zero on the current 1-month OIS by construction. A shock to this factor increases short-term yields on average and the effect diminishes at the longer end of the yield curve.

3.2 Validity of the factors as instruments for monetary policy shocks

Many commonly used HF financial market surprises, similarly extracted from central bank communication events, have been found to be predictable, autocorrelated and also correlated with the central bank's own macroeconomic forecasts. It is therefore reasonable to question whether or not the three identified factors constitute valid instruments for shocks to Riksbank monetary policy. One commonly used indicator of instrument validity in this context is the reaction of asset prices that takes place in the same narrow time window around central bank communications. Following Jarociński and Karadi (2020), several recent studies have focused on the movement of stock market prices. According to economic theory, the stock market should weaken following a monetary policy tightening. However, if the central bank has more information about the state of the economy than the public, and this information is released simultaneously with the monetary policy decision, the resulting monetary policy surprise might move the stock market in the opposite direction. As discussed in the introduction, this channel is sometimes called the Fed information effect. It is not evident to us, however, that prices listed on the Stockholm stock exchange should be expected to react to Riksbank monetary policy shocks in the same way as New York stock prices react to announcements from the Federal Reserve. The Nasdaq Stockholm stock market all share index is heavily influenced by a number of large, multinational corporations who derive most of their revenue from markets outside of Sweden.

Another possibility would be to focus on the sign in the reaction of the krona exchange rate, an asset that can be expected to move in a predictable way to surprises in the Riksbank's announcements. If a tightening (loosening) surprise is associated with an immediate depreciation (appreciation) of the krona, one might suspect that the surprise in question was not caused by a true monetary policy shock. It is not clear to us, however, that such a conclusion would be valid, given the multi-dimensional nature of Riksbank communications, as analysed in the previous section. An unexpected reaction in the exchange rate could be due to, for example, a combination of a target and a path factor shock affecting the market simultaneously. We find some evidence in support of this view, as discussed in Appendix E.

Laséen (2020) investigates the relevance of a Riksbank information effect when estimating the effects of monetary policy in Sweden. Using the Riksbank's own macroeconomic projections, including projections of the policy interest rate, Laséen (2020) constructs an informationally-robust instrument for Swedish monetary policy shocks. The results suggest that in the case of Riksbank announcements, the information channel is not important enough to materially affect the estimated effects of monetary policy.

The other explanation put forth in the literature by Bauer and Swanson (2023a, 2023b), instead focuses on a 'Fed-response-to-news' channel, where market participants continually learn about the central bank's reaction function, as it responds to incoming macroeconomic news. Bauer and Swanson (2023a) show that under the realistic assumption that market participants have only imperfect information about the central bank's reaction function, i.e. about the systematic part of monetary policy, HF financial market surprises may contain two separate components, only one of which is due to true monetary policy shocks. In the simple model of incomplete information proposed by Bauer and Swansson (2023a), the central bank's reaction function is time-varying and not directly observed by market participants. Market participants therefore have to infer information about the reaction function from the central bank's actions, as it responds to macroeconomic news. The second component of financial market surprises is due to market participants updating their knowledge – essentially learning – about the central bank's reaction function.

Bauer and Swanson (2023a) argue that when estimating the effects of monetary policy, it is important to disentangle the two components of financial market surprises and to only use, as an instrument, the component that is due to true monetary policy shocks. The second component, the updating process, is by nature correlated with macroeconomic news and is therefore likely to bias any estimate of the effects of monetary policy on macroeconomic variables. Bauer and Swanson (2023a) show exactly this in their simple model of incomplete information: market surprises may be correlated *ex post* with information that is publicly available prior to the central bank announcement. As a robustness exercise, we make an orthogonalization exercise similar to Bauer and Swanson (2023a) when estimating the effects on the macroeconomy in section 4.

4 Using the identified monetary policy shocks to measure effects on economic activity and inflation

In the previous section, we described how the monetary policy surprises were transformed into a measure of monetary policy shocks using a factor model. We identified three different factors, representing the multi-dimensional nature of monetary policy. In this section, we use the first two factors, the target factor and the path factor, to measure the causal effect of monetary policy on the price level and economic activity in Sweden.^{19, 20}

We follow Gertler and Karadi (2014), and use a proxy VAR in the estimation. In other words, we first estimate the dynamic relationship between the macroeconomic variables in a VAR. We then use the factors as proxies for monetary policy shocks, and estimate the immediate impact effect of these shocks on the macroeconomic variables by regressing the two factors on the reduced form residuals from the VAR estimation.

An alternative to the selected estimation approach would be to either use a local projection IV approach, which directly estimates the effect of the shock on the variable of interest. Another alternative would be to insert the factors as the first variables in a SVAR and use Cholesky identification.²¹ Our selected method does however have one important advantage compared to these methods. We have macroeconomic data on monthly frequency that goes back to January 2000. Our shock sample is however shorter. It only stretches back to August 2004. With the selected approach, we estimate the dynamic relationship between the macroeconomic variables separately from the direct impact of the shocks. This means that we can exploit the full sample, 2000m1-2024m6, when estimating the dynamic relationship, while we use a shorter sample, 2004m8-2024m6, in the estimation of the impact effect.

In addition to using Swedish macroeconomic variables in the estimation, we also control for the foreign economy by including trade-weighted foreign variables as exogenous variables. As Sweden is a small open economy, it is reasonable to assume that changes in economic activity in Sweden have no impact on the global economy. Furthermore, by treating the foreign economy as exogenous, we gain degrees of freedom in the estimation, which is important given our limited sample. The estimated model can be described as:

$$A_0Y_t = C_t + A_1Y_{t-1} + \dots + A_MY_{t-M} + B_0X_t + \dots + B_MX_{t-M} + \varepsilon_t$$
(1)

¹⁹ See Stock and Watson (2018) for a discussion about using instruments to identify causal effects in dynamic macroeconomics. ²⁰ Since the third factor is defined to capture quantitative measures, we can interpret the second factor solely as changes in the

path of the policy rate, excluding any other measures that might change the long end of the yield curve. ²¹ It might be worth pointing out that Plagborg-Møller and Wolf (2021) show that SVAR estimation methods are asymptotically equivalent to local projections.

where Y_t contains the endogenous macroeconomic variables, X_t contains the exogenous macroeconomic variables, and ε_t contains the structural shocks. C_t includes a constant and some dummies, explained later. We first estimate the reduced form VAR relationship and then regress the instruments on the reduced form residuals u_t :

$$u_t = A_0^{-1} \varepsilon_t, \tag{2}$$

using OLS in both estimation procedures. Under the assumption that the monetary policy shocks are orthogonal to other shocks, it is enough to include only a monetary policy shock in the second stage regression, omitting all other structural shocks. Furthermore, the two factors are orthogonal to each other by construction, implying that we can separately estimate also their impact effects.

We set the number of lags to M = 9. The most common choice in the literature is the use of twelve lags, as is the case in for example Gertler and Karadi (2014) as well as Bauer and Swanson (2023b). There are, however, examples of studies using shorter lag lengths, for example Brennan et al. (2024) who use eight lags, and Cesa-Bianchi et al. (2020) use two.²² The reason for choosing nine instead of twelve is to gain degrees of freedom, which is important since our sample is relatively short. As a robustness exercise, we include results with M = 6 and M = 12 in Appendix G.

The macroeconomic data in the estimation spans the period 2000m1–2024m6. We use data in levels, following Sims et al. (1990), Cesa-Bianchi et al. (2020) and Braun et al. (2024). We aggregate the intraday shock data into monthly data by summing all shocks over each month. When estimating the effects of the target factor, we include six endogenous variables in the model: the Swedish policy rate (end-of-period), a measure of monthly GDP distributed by Statistics Sweden (in log level), the unemployment rate for ages 16–64, the CPIF price index (in log level), the real effective exchange rate (in log level), and a measure of financial conditions.²³ Financial conditions are measured by the financial conditions index described in Alsterlind et al. (2020). The financial conditions index is adjusted to accommodate the fact that some of the variables in the index are already part of the VAR.²⁴ When we estimate the effects

²² The authors of the latter study choose their lag length based on an information criterion.

²³ The CPIF is a measure of prices where the interest rate is held constant over time in the index. This is the target variable for the Riksbank. Monthly GDP is also called the *GDP indicator* by statistics Sweden. We measure the monthly policy rate as the end-of-month interest rate level to be consistent with the aggregation method of the monetary policy shocks, in which all shocks over the month are summed up. Another shock aggregation method would be to average the shocks over the month. This would however induce autocorrelation in the shocks (due to parts of the shock spilling over to the next month), which is not consistent with the nature of structural shocks.

²⁴ The financial conditions index consists of a combination of five different indicators: one that captures the housing market, one that captures the stock market, one that captures the bond market, one that captures the money market, and one that captures the foreign exchange market. The last two are removed from the index used in the VAR while the three remaining factors are included

of the path factor on the macroeconomic variables, we add an additional endogenous variable to the estimation: the 2-year swap rate. See Appendix F for a descriptions and sources of the data used in the model.

The foreign variables included in the estimation are the following: a global energy price index, constructed by the IMF, the foreign policy rate, the foreign price level, and foreign industrial production.²⁵ We use industrial production instead of GDP due to data limitations, but also because investment goods have a higher weight in Swedish exports compared to consumption goods.²⁶ In addition to the foreign time series variables, we also include dummies for the pandemic in our estimation, following Bisgaard Larsen and Jessen Weissert (2024), who in turn based their decision to include such dummies on Lenza and Primiceri (2022). We include three dummies in the model, for March, April, and May 2020. The pandemic did not significantly affect the Swedish economy until March 2020, when restrictions were put in place to limit the spread of the disease.²⁷

The F-statistics for the two factors as instruments are 27.8 for the estimation with the target factor, and 7.5 for the path factor. Hence, the path factor instrument is a bit weak, relative to the threshold of 10 suggested by Stock and Watson (2012). The F-statistic for the target factor is well above 10, indicating a strong instrument.²⁸ We calculate the confidence bands around the point estimates using bootstrapping techniques, following Amberg et al. (2022), who use the moving blocks bootstrap technique recommended by Jentsch and Lunsford (2019).

4.1 The effect of monetary policy shocks on the macroeconomy

Figure 4 illustrates the effects of a *target shock* on the macroeconomic variables. The blue line shows the point estimates of the effects. The blue field shows the 68 per cent confidence interval and the dashed blue lines show the outer bounds of the 90 per cent confidence intervals. The shock is normalized such that the impact effect on the policy rate is 1 percentage point.

in the index with equal weights. The reason for the removal of the money market indicator is that it includes the three month STIBOR rate, a variable tightly linked to the Swedish policy rate. The policy rate is already included in the VAR. The reason for why the foreign exchange market is removed is that it includes the nominal exchange rate, which is also included in the VAR.

²⁵ The foreign economy is represented by a trade-weighted average of the US and the Euro Area, of which the Euro Area carries the highest weight.

²⁶ Indirect evidence of this can be found in Corbo and Strid (2020), who estimate a DSGE model of the Swedish and foreign economy and finds a weight of 0.73 on foreign investment and 0.27 on foreign consumption in their export demand function.

²⁷ For more information about the Swedish response to the Covid-19-pandemic and its effect on the economy, see for example National Institute of Economic Research (2020).

 $^{^{28}}$ The F-test is constructed by estimating a regression with the reduced form residual of the target variable (the policy rate for the target factor and the 2-year swap rate for the path factor) on the left-hand-side and its respective instrument together with *M* lags of the reduced form residual on the right-hand-side, then squaring the t-value of the instrument coefficient. The t-values are calculated using MacKinnon and White (1985) robust standard errors.

All impulse responses have the expected signs, with the effect on GDP and the price level being quite small on impact. The effect on the policy rate is transient, but quite persistent, being above zero for about two years. Both financial conditions and the exchange rate react quickly. Financial conditions tighten while the exchange rate appreciates. Both of these results are as expected. Financial conditions capture housing prices, the stock market and a credit spread. A higher policy rate should, ceteris paribus, lead to lower asset prices (lower housing prices and a lower stock market index), and hence to tighter financial conditions. The effect on the real exchange rate is strong, stronger than what is implied by a regular UIP calculation. Furthermore, it takes a hump shaped form, in line with previous studies (see for example Lindé et al. (2009)).²⁹



Figure 4: Effect on macroeconomic variables of a target factor shock

Note: Data is monthly. The policy rate is measured at its end-of-period value. The solid lines show the point estimates of the impulse response functions following a target factor shock. The impulse responses are normalised to yield a 1 pp. higher policy rate on impact. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured in per cent deviations.

The point estimate of GDP during the first months after the shock is close to zero. The effect does however increase over time. GDP reaches its trough after more than two years. Hence, the impulse response implies a sustained period of sluggish growth after a target rate shock. As is the case for GDP, it takes some time for the price level to react in any economically meaningful way. However, once it has started to decline, it does so until about two years after the shock. The trough of the point estimate of the price level is close to -1 per cent, after which

²⁹ Furthermore, impact value on the real exchange rate is well in-line with a regression on intraday data with the shock on the right-hand-side and the EUR/SEK exchange rate on the left-hand-side.

it starts to increase again. The maximum effect on the price level is about 50-60 per cent of the maximum effect on the level of GDP. In contrast, the increase in unemployment arrives faster. The effect on unemployment is small; the unemployment rate increases somewhat in the months following the shock and continues to be on slightly elevated levels for about two years.

We now turn to the effect of a *path factor shock* on the macroeconomic variables. The shock is normalized such that the 2-year swap rate increases by 1 percentage point on impact, representing credible but unexpected communication by the central bank that monetary policy will be tighter in the future. Different central banks use different communication strategies regarding the future path of the policy rate. In the case of the Riksbank, such communication consists of a combination of a published interest rate path together with written communication. Our identification scheme does however not allow us to separate the effects of written communication from the published policy rate path.³⁰

The effects of a path factor shock are illustrated in Figure 5. As shown in the figure, there is a relatively fast downward correction of the swap rate towards zero. It does however stay at elevated levels for about two years.

³⁰ This is in contrast to Natvik et al. (2020), who explicitly investigate how changes in the published policy rate forecast affects the expectations of future policy rates by the market.



Figure 5: Effect on macroeconomic variables of a path factor shock

Note: Data is monthly. The policy rate and the 2-year swap rate are measured at their end-of-period values. The solid lines show the point estimates of the impulse response functions following a path factor shock. The impulse responses are normalized such that the initial impact on the 2-year swap rate is 1 pp. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are all measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured in per cent deviations.

The path factor shock leads to an immediate tightening of financial conditions. However, in contrast to a target factor shock, the point estimate of the real exchange rate does not depreciate immediately following the shock. Instead, the real exchange becomes materially stronger than baseline only after about half a year.

The policy rate increases immediately following the shock, despite the chosen identification strategy. One reason for this could be that we aggregate data from intraday to monthly by taking the sum over the month. After impact, the policy rate continues to increase, reaching its maximum after about one and a half years. It then stays at elevated levels for another year and a half.

The effect on GDP is similar to that of a target factor shock. This is also the case for the price level, except that it takes a longer time for the price level to start to decrease. The ratio between the maximum effect of the point estimate on the price level and the maximum point estimate effect of the level of GDP is around 50 per cent, somewhat lower than that for the target factor shock.

We now proceed by comparing our results to those of some other studies that also use high frequency methods to identify Swedish monetary policy shocks. Amberg et al. (2022) use daily shock data from financial markets, and find smaller effects on inflation than we do. Their

sample does however end in 2018m12. In Appendix G, we show results from a robustness exercise where we re-estimate our model on a shorter sample, ending in 2019m12. That exercise yields a smaller effect on prices than our baseline estimation, though it is still somewhat stronger than those of Amberg et al. (2022). Another study that uses similar methods is Laséen (2020), who also has a shorter sample period relative ours. He also finds a smaller effect on the price level in the short run compared to us. The dynamics of the price index do differ, however, with a more persistent decrease in Laséen (2020). Laséen and Nilsson (2024) use a sample period that is more similar to ours, ending in 2023m8. In contrast to the previously mentioned studies, they find stronger effects than we do. Their identification strategy is however slightly different, and they identify more persistent changes in the policy interest rate, compared to us. This could explain their relatively strong effect on prices.³¹

How do our results compare to the results estimated for other small open economies? One study which is methodologically close to ours is Braun et al. (2024), reporting results for both a target factor shock and a path factor shock. Their results differ somewhat from ours. The policy rate in the case of a target factor shock reverts faster down towards zero in their study. Their effects on prices from a target rate shock are also smaller than what we find. The maximum effect does however arrive around the same time in both studies. The magnitude of the effect on GDP is more similar to our study, but the maximum effect arrives earlier.

Regarding the path factor shock in Braun et al. (2024), they report a fairly quick response on GDP, which then reverts back to zero relatively quickly. In the case of the price level, Braun et al. (2024) find that it is continuously decreasing over the whole period represented in their graphs; according to our estimates, the price level instead reaches a trough after around two years.

Another study related to the effects on small open economies is Soosalu (2024), who uses a HF change of the Canadian three-month contract BAX as an instrument for Canadian monetary policy shocks. Soosalu (2024) finds a persistent effect on both GDP and prices, just as we do. The relationship ratio between the maximum effect of prices and GDP is also quite close to the one in our study.

An additional study for small open economies is Bisgaard Larsen and Jessen Weissert (2024) who find larger effects on the economy compared to us, using local projections. One reason for this could be that they are studying an economy with a fixed exchange rate, looking at the

³¹ A recent study that looks at Swedish data, but that uses a standard Cholesky identification scheme instead of HF methods are Berggren et al. (2024) who find effects on prices that are fairly similar to ours.

effects of euro area monetary policy shocks on economic outcomes in Denmark. Euro area monetary policy shocks also have an effect on the European economy, which spills over on the Danish economy, leading to stronger effects.

4.2 Robustness

We now continue by investigating how our results change if we change certain assumptions and adjust the number of observations when we construct the monetary policy shocks. In Appendix G, we also investigate how our results are affected by changes in the specification of the VAR model, finding that our main results are robust to several such changes.

In this section, however, we focus on robustness to the specification of the monetary policy shocks. Taking into consideration the robustness exercises, including those in the Appendix, we can draw the following conclusions: i) all variables are qualitatively behaving according to theoretical predictions following a monetary policy shock; there is no price puzzle in our estimations, as is frequently found in the literature, and ii) GDP and prices respond with a lag. The maximum effect on GDP and the price level comes after more than one year but less than three years. Finally, iii) unemployment responds quickly, and less than GDP.

A common concern when using instruments in estimations is their relevance and exogeneity. The relevance criterion is usually handled by measures of instrument strength (we use the F-statistic to measure the strength of the instrument, as is standard in the literature). To ensure exogeneity, different approaches are used in the literature. Furthermore, different hypotheses are introduced in the literature to why monetary policy surprises might not reflect true monetary policy shocks. Two such hypotheses are the Fed information effect and the Fed responding to news channel. The two effects are discussed in more detail in Section 3. Laséen (2020) investigates if central bank information shocks affect the results when estimating effects of Swedish monetary policy shocks, finding no evidence for this. Bauer and Swanson (2023a, 2023b) introduces the Fed responding to news channel as a possible concern for monetary policy shocks, and proposes a way to clean the monetary policy surprises from any such effects.

As a first robustness exercise, we follow their suggestion and orthogonalize the monetary policy surprises by regressing them on real time data that is correlated with the monetary policy surprises and that were publicly available on the day prior to each monetary policy announcement. We then use the residuals from that regression as the cleaned monetary policy surprises. Appendix C contains more details on the cleaning procedure, and also contains a table with correlations which we use to determine which variables should be included in the

cleaning procedure. The results from this exercise are shown in Figure 6. Concerning shocks to the target factor, the initial response of the real exchange rate is somewhat weaker when orthogonalized surprises are used as instruments, compared to the baseline results. The effect is however still quite large, and over time the point estimate converges towards that of the baseline estimation. Financial conditions and the unemployment rate reacts somewhat stronger in the first 6 to 12 months following the shock, but the results are otherwise similar to baseline. For the path factor, the maximum effects on both inflation and GDP are weaker, but the results are qualitatively similar. We conclude that imperfect knowledge of the true parameters of the Riksbank's reaction function does not seem to systematically affect estimates of the macroeconomic transmission of monetary policy shocks in Sweden.³²

Second, we investigate how the results are altered when we use monetary policy shocks that are based on monetary policy surprises constructed using daily data, in contrast to the intraday data we use in the baseline estimation. The resulting IRFs are shown in Figure 7.³³ The dashed red line shows the point estimate from the baseline estimation. As shown in the figure, the impact effect on financial conditions and the real exchange rate are stronger in the case of a target factor shock. This does in turn lead the maximum effects of the price level and GDP to somewhat stronger.

The impact effect on financial conditions for the path factor estimation is instead substantially weaker than in the baseline case. Furthermore, the dynamics of the policy rate differ significantly, the maximum effect arriving quickly after the initial shock. There is also a substantial change in the path of unemployment, GDP and the exchange rate in this exercise. One possible conclusion is that it is more important to use intraday data when estimating effects of shocks to the path factor, compared to estimations of shocks to the target factor. As is illustrated in Section 2, the daily movements in Swedish interest rate instruments with longer maturities are highly correlated with international interest rates, a fact that may introduce noise into the estimation.

In a third robustness exercise, reported in Figure 8, we investigate how the exclusion of minutes and extraordinary meetings affect the results. The target factor instrument is strong, with an F-statistic above 30, while for the path factor, the instrument is fairly weak, with an F-statistic of 5.4. This suggests that the inclusion of extraordinary meetings and minutes are important for the strength of the path factor. The results for the target shock are practically

³² The F-test statistic of the target factor and the path factor in the exercises are 15.1 and 9.6 respectively, suggesting that both instruments are strong enough, even if the path factor lies slightly below the threshold of 10.

³³ The F-test statistics of the target factor and the path factor in the exercises are 17.0 and 15.4 respectively.

unchanged compared to the baseline estimation, while the differences are somewhat larger for the path factor shock. This is however to be expected. The minutes do not contain any new information about the current policy rate, since they are related to policy rate decisions that have already been announced. However, the minutes are likely to contain information about possible future changes in the monetary policy stance, suggesting that adding surprises related to the release of minutes be important for the results of the path factor shock. In general, the macroeconomic results following a path factor shock are qualitatively similar to the baseline case, but stronger when minutes and extraordinary meetings are excluded.



Figure 6: Robustness – effect on macroeconomic variables when surprises are orthogonalized

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.



Figure 7: Robustness – effect on macroeconomic variables when daily shock data is used

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Figure 8: Robustness – effect on macroeconomic variables when only regular meetings are included



Panel A: effect of target factor shock

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

5 Conclusions

What are the effects of monetary policy on key macroeconomic variables such as output, inflation and unemployment? This paper attempts to answer this question by estimating the macroeconomic effects of monetary policy shocks in Sweden using a novel monetary policy event study database with high-frequency financial market reactions to Riksbank communications, spanning a period of 20 years. These reactions are constructed using intraday tick data for seven different swap instruments that reflect expected interest rates at different horizons: the 1-month, 3-month and 6-month SEK overnight index swap (STINA) rates, as well as the 1-year, 2-year, 5-year and 10-year swap rates. We follow a simple event study approach where we observe how the swap rates change around a 30-minute window at the time of Riksbank monetary policy communication events.

Following the methodology of Gürkaynak et al. (2005b) and Swanson (2021), we decompose the estimated reactions and identify three types of monetary policy shocks: shocks to the current level of the policy rate, shocks to its expected future level and shocks related to the Riksbank's use of quantitative monetary policy measures. We focus on the first two of these types of unexpected changes in Riksbank monetary policy and use them as external instruments in a proxy VAR-model to estimate their effects on key macroeconomic variables. We find that an unexpected, temporary tightening (loosening) of the policy rate induces an immediate and persistent appreciation (depreciation) of the krona exchange rate, as well as a gradual, persistent, negative (positive) effect on output and consumer prices. These effects are statistically significant, economically meaningful and robust to a number of variations in our econometric specification. We also estimate the effects of shocks to market participants' expectations about the future level of the policy interest rate. The effects on output and consumer prices are similar to those of an unexpected, temporary change in the current level of the policy rate. However, in the case of shocks to the expected future level of the policy rate, the instrument is on the weak side, and the results are less robust.

To conclude, the contribution of this paper is threefold. First, we make available a new event study database with intraday financial market reactions to Riksbank communications, covering both regular and extraordinary monetary policy announcements as well as minutes of monetary policy meetings. This more than doubles the sample size available to identify monetary policy shocks compared to previous Swedish studies. The database hence provides a valuable resource for researchers and we hope that it will facilitate more research on the effects of monetary policy in Sweden. Second, using this database, we are able to decompose the monetary policy surprises into three different types of monetary policy shocks: shocks to

the current level of the policy rate, shocks to its expected future level and shocks related to the Riksbank's use of quantitative monetary policy measures. Third, we make use of these shocks to estimate the effects on key macroeconomic variables, contributing to a better understanding of how monetary policy transmits through the Swedish economy.

6 References

Alsterlind, J., Lindskog, M., and von Brömsen, T. (2020). An index for financial conditions in Sweden. *Staff memo*, Sveriges Riksbank.

Altavilla, C., Brugnolini, L., Gürkaynak, R. S., Motto, R., and Ragusa, G. (2019). Measuring euro area monetary policy. *Journal of Monetary Economics*, 108(C):162--179.

Amberg, N., Jansson, T., Klein, M., and Rogantini Picco, A. (2022). Five facts about the distributional income effects of monetary policy shocks. *American Economic Review: Insights*, 4(3):289--304.Bauer, M. D. and Swanson, E. T. (2023a). An alternative explanation for the "Fed information effect". *American Economic Review*, 113(3):664--700.

Bauer, M. D. and Swanson, E. T. (2023b). A Reassessment of Monetary Policy Surprises and High-Frequency Identification. *NBER Macroeconomics Annual*, vol. 37(1):87-155.

Berggren, E., Mammos, S. and Strid, I. (2024). The effects of monetary policy in Sweden during the inflation targeting period: estimates with structural VAR models, *Staff memo*, Sveriges Riksbank.

Bisgaard Larsen, R. and Jessen Weissert, C. (2024). Monetary policy transmission in Denmark. *Working paper*, Danmarks nationalbank.

Braun, R., Miranda-Agrippino, S., and Saha, T. (2024). Measuring monetary policy in the UK: the UK Monetary Policy Event-Study Database. *Journal of Monetary Economics*, 103645.

Brennan, Connor M., Margaret M. Jacobson, Christian Matthes, and Todd B. Walker (2024). Monetary Policy Shocks: Data or Methods?, *Finance and Economics Discussion Series* 2024-011. Washington: Board of Governors of the Federal Reserve System

Brubakk, L., ter Ellen, S., Robstad, r., and Xu, H. (2022). The macroeconomic effects of forward communication. *Journal of International Money and Finance*, 120(C).

Brubakk, L., ter Ellen, S., and Xu, H. (2021). Central bank communication through interest rate projections. *Journal of Banking & Finance*, 124(C).

Campbell, J. R., Evans, C. L., Fisher, J. D., Justiniano, A., Calomiris, C. W., & Woodford, M. (2012). Macroeconomic effects of federal reserve forward guidance. *Brookings papers on economic activity*, 1-80.

Cesa-Bianchi, A., Thwaites, G., and Vicondoa, A. (2020). Monetary policy transmission in the United Kingdom: A high frequency identification approach. *European Economic Review*, 123:103375.

Corbo, V. and Strid, I. (2020). MAJA: A two-region DSGE model for Sweden and its main trading partners. *Working Paper Series* 391, Sveriges Riksbank

Cragg, John G. & Donald, Stephen G. (1997). Inferring the rank of a matrix. Journal of Econometrics, vol. 76(1-2):223-250.

De Rezende, R. B. (2017). How can term structure models be used by central banks?, Sveriges Riksbank Economic Review

Ellingsen, T., & Söderström, U. (2001). Monetary policy and market interest rates. *American Economic Review*, 91(5), 1594-1607.

Faust, J., J. H. Rogers, E. T. Swanson, and J. H. Wright. 2003. Identifying the Effects of Monetary Policy Shocks on Exchange Rates using High-Frequency Data. *Journal of the European Economic Association* 1(5): 1031–57

Fransson, L. and Tysklind, O. (2016). What effect does monetary policy have on interest rates? *Sveriges riksbank Economic Review*, 2016(1):38--60.

Gertler, M. and Karadi, P. (2014). Monetary policy surprises, credit costs, and economic activity. *American Economic Journal: Macroeconomics*, 7(1):44--76.

Gürkaynak, R., Sack, B., and Swanson, E. (2005a). The sensitivity of long-term interest rates to economic news: Evidence and implications for macroeconomic models. *American Economic Review*, 95(1):425--436.

Gürkaynak, R. S., Sack, B., and Swanson, E. (2005b). Do Actions Speak Louder Than Words? The Response of Asset Prices to Monetary Policy Actions and Statements. *International Journal of Central Banking*, 1(1).

Hambur, J. and Haque, Q. (2023). Can We Use High-frequency Yield Data to Better Understand the Effects of Monetary Policy and Its Communication? Yes and No! *RBA Research Discussion Papers* 2023-04, Reserve Bank of Australia.

Iversen, J. and Tysklind, O. (2017). The effect of repo rate path changes on asset prices. *Staff memo*, Sveriges Riksbank.

Jarociński, M., & Karadi, P. (2020). Deconstructing monetary policy surprises—the role of information shocks. *American Economic Journal: Macroeconomics*, 12(2), 1-43.

Jentsch, C. and Lunsford, K. G. (2019). The dynamic effects of personal and corporate income tax changes in the United States: Comment. *American Economic Review*, 109(7):2655--78.

Kilman, J. (2022). Monetary policy shocks for Sweden. Working paper 2022:18, Lund University.

Kuttner, K. (2001). Monetary policy surprises and interest rates: Evidence from the fed funds futures market. *Journal of Monetary Economics*, 47(3):523--544.

Laséen, S. (2020). Monetary Policy Surprises, Central Bank Information Shocks, and Economic Activity in a Small Open Economy. *Working Paper Series 396*, Sveriges Riksbank

Laséen, S. and Nilsson, C. (2024). How does the Riksbank's monetary policy affect the Swedish economy: does inflation rise when the policy rate is raised?. *Staff memo*, Sveriges Riksbank.

Lenza, M. and Primiceri, G. E. (2022). How to estimate a vector autoregression after March 2020. *Journal of Applied Econometrics*, 37(4):688--699.

Lindé, J., Nessén, M., and Söderström, U. (2009). Monetary policy in an estimated open-economy model with imperfect pass-through. *International Journal of Finance & Economics*, 14(4):301--333.

MacKinnon, J. G. and White, H. (1985). Some heteroskedasticity-consistent covariance matrix estimators with improved finite sample properties. *Journal of Econometrics*, 29(3):305--325.

Miranda-Agrippino, S., & Ricco, G. (2021). The transmission of monetary policy shocks. American Economic Journal: Macroeconomics, 13(3), 74-107.

Nakamura, E. & Steinsson, J. (2018). High-Frequency Identification of Monetary Non-Neutrality: The Information Effect. *The Quarterly Journal of Economics*, vol. 133(3):1283-1330.

Natvik, G. J., Rime, D., and Syrstad, O. (2020). Does publication of interest rate paths provide guidance? *Journal of International Money and Finance*, 103(C).

National Institute of Economic Research, N. I. (2020). Makroekonomiska och samhällsekonomiska effekter av de vidtagna åtgärderna för att dämpa spridningen av covid-19 i Sverige. *Occasional study*, KI 2020:25, National Institute of Economic Research.

Plagborg-Møller, M. and Wolf, C. K. (2021). Local Projections and VARs Estimate the Same Impulse Responses. *Econometrica*, 89(2):955--980.

Romer, C. D., & Romer, D. H. (2000). Federal Reserve information and the behavior of interest rates. *American* economic review, 90(3), 429-457.

Sandström, M. (2018). The impact of monetary policy on household borrowing - a high-frequency IV identification. *Working Paper Series 351*, Sveriges Riksbank

Sims, C.A., Stock, J.H. and Watson, M.W. (1990). Inference in linear time series models with some unit roots. *Econometrica* 58 (1):113–144

Soosalu, Matt. (2024), Monetary Policy Transmission in Canada – A High Frequency Identification Approach, *The B.E. Journal of Macroeconomics*, 2024.

Stock, J. H. and Watson, M. W. (2012). Disentangling the Channels of the 2007-09 Recession. *Brookings Papers on Economic Activity*, Economic Studies Program, The Brookings Institution, vol. 43(1): 81--156.

Stock, J. H. and Watson, M. W. (2018). Identification and estimation of dynamic causal effects in macroeconomics using external instruments. *The Economic Journal*, 128(610):917--948.

Sveriges Riksbank (2023a). The new Riksbank Act and the monetary policy framework, article in *Monetary Policy Report*, February 2023.

Sveriges Riksbank (2023b). Monetary Policy Report, September 2023.

Swanson, E. T. (2021). Measuring the effects of Federal Reserve forward guidance and asset purchases on financial markets. *Journal of Monetary Economics*, 118(C):32--53.

Åhl, M. (2017). How big is the toolbox of a central banker? Managing expectations with policy-rate forecasts: Evidence from Sweden. *Working Paper Series* 339, Sveriges Riksbank

APPENDIX A

The table shows the Riksbank communication events that we use in the analysis. This differs by four dates compared to the published database. See footnote 14 in section 3 for details on the dates that we exclude and why.

Date	Type of Riksbank announcement	Time
2004-08-20	Regular Announcement	09:30 CET
2004-09-02	Minutes	09:30 CET
2004-10-14	Regular Announcement	08:00 CET
2004-10-27	Minutes	09:30 CET
2004-12-09	Regular Announcement	09:30 CET
2004-12-22	Minutes	09:30 CET
2005-01-28	Regular Announcement	09:30 CET
2005-02-10	Minutes	09:30 CET
2005-03-15	Regular Announcement	09:00 CET
2005-04-01	Minutes	09:30 CET
2005-04-29	Regular Announcement	09:30 CET
2005-05-17	Minutes	09:30 CET
2005-06-21	Regular Announcement	09:30 CET
2005-07-05	Minutes	09:30 CET
2005-08-24	Regular Announcement	09:30 CET
2005-09-12	Minutes	09:30 CET
2005-10-20	Regular Announcement	08:00 CET
2005-10-31	Minutes	09:30 CET
2005-12-02	Regular Announcement	09:30 CET
2005-12-15	Minutes	09:30 CET
2006-01-20	Regular Announcement	09:30 CET
2006-02-02	Minutes	09:30 CET
2006-02-23	Regular Announcement	08:00 CET
2006-03-08	Minutes	09:30 CET
2006-04-28	Regular Announcement	09:30 CET
2006-05-11	Minutes	09:30 CET
2006-06-20	Regular Announcement	09:30 CET
2006-07-03	Minutes	09:30 CET
2006-08-30	Regular Announcement	09:30 CET
2006-09-12	Minutes	09:30 CET
2006-10-26	Regular Announcement	08:00 CET
2006-11-15	Minutes	09:30 CET
2006-12-15	Regular Announcement	09:30 CET
2007-01-02	Minutes	09:30 CET
2007-02-15	Regular Announcement	09:30 CET
2007-02-28	Minutes	09:30 CET
2007-03-30	Regular Announcement	09:30 CET
2007-04-16	Minutes	09:30 CET
2007-05-04	Regular Announcement	09:30 CET
2007-05-18	Minutes	09:30 CET
2007-06-20	Regular Announcement	09:30 CET
2007-07-04	Minutes	09:30 CET
2007-09-07	Regular Announcement	09:30 CET
2007-09-20	Minutes	09:30 CET
2007-10-30	Regular Announcement	09:30 CET
2007-11-13	Minutes	09:30 CET
2007-12-19	Regular Announcement	09:30 CET
2008-01-08	Minutes	09:30 CET
2008-02-13	Regular Announcement	09:30 CET
2008-02-25	Minutes	09:30 CET
2008-04-23	Regular Announcement	09:30 CET
2008-05-07	Minutes	09:30 CET
2008-07-03	Regular Announcement	09:30 CET
2008-07-16	Minutes	09:30 CET
2008-09-04	Regular Announcement	09:30 CET
2008-09-17	Minutes	09:30 CET
2008-10-02	Extraordinary Announcement	09:30 CET
2008-10-06	Extraordinary Announcement	09:30 CET
2008-10-13	Extraordinary Announcement	09:30 CET
2008-10-23	Regular Announcement	09:30 CET
2008-10-24	Extraordinary Announcement	09:30 CET
2008-10-29	Extraordinary Announcement	09:30 CET
2008-11-05	Minutes	09:30 CET
2008-12-01	Extraordinary Announcement	09:30 CET
2008-12-04	Regular Announcement	09:30 CET

2008-12-17	Minutes	09:30 CET
2009-02-11	Regular Announcement	09:30 CET
2009-02-13	Extraordinary Announcement	09:30 CET
2009-02-25	Minutes	09:30 CET
2009-04-21	Regular Announcement	09:30 CET
2009-05-05	Minutes	09:30 CET
2009-05-07	Extraordinary Announcement	09:30 CET
2009-07-02	Regular Announcement	09:30 CET
2009-07-16	Minutes	09:30 CET
2009-09-03	Regular Announcement	09:30 CET
2009-09-16	Minutes	09:30 CET
2009-10-22	Regular Announcement	09:30 CET
2009-11-04	Minutues	09:30 CET
2009-11-05	Extraordinary Announcement	09:30 CET
2009-12-16	Regular Announcement	09:30 CET
2010-01-04	Minutes	09:30 CET
2010-02-11	Regular Announcement	09:30 CET
2010-02-24	Minutes	09:30 CET
2010-04-20	Regular Appouncement	09:30 CET
2010-05-03	Minutes	09:30 CET
2010-07-01	Regular Announcement	09:30 CET
2010-07-15	Minutes	09:30 CET
2010-09-02	Regular Announcement	09:30 CET
2010-09-15	Minutes	09:30 CET
2010-10-26	Regular Announcement	09:30 CET
2010-11-08	Minutes	09:30 CET
2010-12-15	Regular Announcement	09:30 CET
2011-01-03	Minutes	09:30 CET
2011-02-15	Regular Announcement	09:30 CET
2011-02-28	Minutes	09:30 CET
2011-04-20	Regular Announcement	09:30 CET
2011-05-03	Minutes	09:30 CET
2011-07-05	Regular Announcement	09:30 CET
2011-07-18	Minutes	09:30 CET
2011-09-07	Regular Announcement	09:30 CET
2011-09-20	Minutes	09:30 CET
2011-10-27	Regular Announcement	09:30 CET
2011-11-09	Minutes	09:30 CET
2011-12-20	Regular Announcement	09:30 CET
2012-01-03	Minutes	09:30 CET
2012-02-16	Regular Announcement	09:30 CET
2012-02-29	Minutes	09:30 CET
2012-04-18	Regular Announcement	09:30 CET
2012-05-02	Minutes	09:30 CET
2012-07-04	Regular Announcement	09:30 CET
2012-07-17	Minutes	09:30 CET
2012-09-06	Regular Announcement	09:30 CET
2012-09-18	Minutes	09:30 CET
2012-10-25	Regular Announcement	09:30 CET
2012-11-07	Minutes	09:30 CET
2012-12-18	Regular Announcement	09:30 CET
2013-01-08	Minutes	09:30 CET
2013-02-13	Regular Announcement	09:30 CET
2013-02-26	Minutes	09:30 CET
2013-04-17	Regular Announcement	09:30 CET
2013-04-29	Minutes	09:30 CET
2013-07-03	Regular Announcement	09:30 CET
2013-07-16	Minutes	09:30 CET
2013-09-05	Regular Announcement	09:30 CET
2013-09-18	Minutes	09:30 CET
2013-10-24	Regular Announcement	09:30 CET
2013-11-06	Minutes	09:30 CET
2013-12-17	Regular Announcement	09:30 CET
2014-01-08	Minutes	09:30 CET
2014-02-13	Regular Announcement	09:30 CET
2014-02-26	Minutes	09:30 CET

2014-04-09	Regular Announcement	09:30 CET
2014-04-24	Minutes	09:30 CET
2014-07-03	Regular Announcement	09:30 CET
2014-07-16	Minutes	09:30 CET
2014-09-04	Kegular Announcement	09:30 CET
2014-09-17	Regular Appouncement	09:30 CET
2014-10-28	Minutes	09.30 CET
2014-12-16	Regular Announcement	09:30 CET
2015-01-08	Minutes	09:30 CET
2015-02-12	Regular Announcement	09:30 CET
2015-02-25	Minutes	09:30 CET
2015-03-18	Extraordinary Announcement	09:30 CET
2015-04-29	Regular Announcement	09:30 CET
2015-05-12	Minutes	09:30 CET
2015-07-02	Regular Announcement	09:30 CET
2015-07-15	Regular Appouncement	09:30 CET
2015-09-05	Minutes	09:30 CET
2015-10-28	Regular Announcement	09:30 CET
2015-11-10	Minutes	09:30 CET
2015-12-15	Regular Announcement	09:30 CET
2016-01-04	Extraordinary Announcement	09:30 CET
2016-01-08	Minutes	09:30 CET
2016-02-11	Regular Announcement	09:30 CET
2016-02-22	Minutes Regular Announcer t	09:30 CET
2016-04-21	Kegular Announcement	09:30 CET
2016-03-04	Regular Announcement	09:30 CET
2016-07-19	Minutes	09:30 CET
2016-09-07	Regular Announcement	09:30 CET
2016-09-21	Minutes	09:30 CET
2016-10-27	Regular Announcement	09:30 CET
2016-11-09	Minutes	09:30 CET
2016-12-21	Regular Announcement	09:30 CET
2017-01-12	Minutes	09:30 CET
2017-02-15	Minutos	09:30 CET
2017-03-01	Regular Announcement	09:30 CET
2017-05-10	Minutes	09:30 CET
2017-07-04	Regular Announcement	09:30 CET
2017-07-18	Minutes	09:30 CET
2017-09-07	Regular Announcement	09:30 CET
2017-09-21	Minutes	09:30 CET
2017-10-26	Regular Announcement	09:30 CET
2017-11-09	Minutes Regular Announcement	09:30 CET
2017-12-20	Regular Announcement	09:30 CET
2018-02-23	Minutes	09:30 CET
2018-04-26	Regular Announcement	09:30 CET
2018-05-08	Minutes	09:30 CET
2018-07-03	Regular Announcement	09:30 CET
2018-07-12	Minutes	09:30 CET
2018-09-06	Regular Announcement	09:30 CET
2018-09-17	Minutes	09:30 CET
2018-10-24	Negular Announcement	09:30 CET
2018-11-02	Regular Announcement	09.30 CET
2019-01-09	Minutes	09:30 CET
2019-02-13	Regular Announcement	09:30 CET
2019-02-22	Minutes	09:30 CET
2019-04-25	Regular Announcement	09:30 CET
2019-05-07	Minutes	09:30 CET
2019-07-03	Regular Announcement	09:30 CET
2019-07-12	Minutes	09:30 CET
2019-09-05	Neguar Announcement	09:30 CET
2019-10-24	Regular Announcement	09:30 CET
2019-11-05	Minutes	09:30 CET
		•

2019-12-19	Regular Announcement	09:30 CET
2020-01-08	Minutes	09:30 CET
2020-02-12	Regular Announcement	09:30 CET
2020-02-21	Minutes	09:30 CET
2020-03-13	Extraordinary Announcement and Minutes	09:30 CET
2020-03-16	Extraordinary Announcement	09:30 CET
2020-03-19	Extraordinary Announcement	09:30 CET
2020-03-26	Extraordinary Announcement and Minutes	09:30 CET
2020-03-27	Extraordinary Announcement	09:30 CET
2020-03-30	Minutes	09:30 CET
2020-04-06	Extraordinary Announcement	09:30 CET
2020-04-07	Minutes	09:30 CET
2020-04-22	Extraordinary Announcement	09:30 CET
2020-04-28	Regular Announcement	09:30 CET
2020-05-04	Minutes	09:30 CET
2020-05-08	Extraordinary Announcement	09:30 CET
2020-05-11	Minutes	09:30 CET
2020-05-15	Extraordinary Announcement	09:30 CET
2020-07-01	Minutos	09:30 CET
2020-07-10	Extraordinary Appouncement	09:30 CET
2020-09-01	Regular Announcement	09:30 CET
2020-10-01	Minutes	09:30 CET
2020-11-26	Regular Announcement	09:30 CET
2020-12-07	Minutes	09:30 CET
2021-02-10	Regular Announcement	09:30 CET
2021-02-19	Minutes	09:30 CET
2021-04-27	Regular Announcement	09:30 CET
2021-05-10	Minutes	09:30 CET
2021-07-01	Regular Announcement	09:30 CET
2021-07-12	Minutes	09:30 CET
2021-09-21	Regular Announcement	09:30 CET
2021-09-30	Minutes	09:30 CET
2021-11-25	Regular Announcement	09:30 CET
2021-12-06	Minutes	09:30 CET
2022-02-10	Regular Announcement	09:30 CET
2022-02-21	Minutes Regular Appouncement	09:30 CET
2022-04-20	Minutes	09:30 CET
2022-06-30	Regular Announcement	09:30 CET
2022-07-11	Minutes	09:30 CET
2022-09-20	Regular Announcement	09:30 CET
2022-09-29	Minutes	09:30 CET
2022-11-24	Regular Announcement	09:30 CET
2022-12-05	Minutes	09:30 CET
2023-02-09	Regular Announcement	09:30 CET
2023-02-20	Minutes	09:30 CET
2023-04-26	Regular Announcement	09:30 CET
2023-05-09	Minutes	09:30 CET
2023-06-29	Regular Announcement	09:30 CET
2023-07-10	Minutes	09:30 CET
2023-09-21	Regular Announcement	09:30 CET
2023-10-02	Minutes	09:30 CET
2023-11-23	Keguar Announcement	09:30 CET
2023-12-04	Regular Appoincement	09:30 CE1
2024-02-01	Minutes	09.30 CE1
2024-02-07	Regular Announcement	09:30 CET
2024-03-27	Minutes	09:30 CET
2024-05-08	Regular Announcement	09:30 CET
2024-05-15	Minutes	09:30 CET
2024-06-27	Regular Announcement	09:30 CET
2024-07-03	Minutes	09:30 CET
2024-08-20	Regular Announcement	09:30 CET
2024-08-26	Minutes	09:30 CET

APPENDIX B

H_0 : number of factors equals	Wald statistic	p-value
0	32.7	0.000
1	23.7	0.001
2	15.5	0.013
3	7.8	0.249

Table B.1 Tests of number of factors characterizing Riksbank announcements

Note: The table shows the results from the Cragg and Donald (1997) matrix rank test. It tests the null hypothesis of m_0 common factors against the alternative of $m > m_0$ common factors.

APPENDIX C

Orthogonalizing the surprises with respect to financial and macroeconomic news

To ensure that our instruments of Swedish monetary policy shocks are not contaminated by markets updating their information concerning the Riksbank's reaction function, we investigate whether or not financial market surprises around Riksbank communications are correlated with information that was publicly available at the time of those announcements. To this end, we construct a real-time set of financial and macroeconomic data covering both domestic and international economic conditions. This real-time data set includes 14 different variables. For each of the 265 observations in our sample of Riksbank communication events, we collect, for each of these 14 variables, the last vintage time series that was publicly available on the day preceding the event. We then compute the change in each variable over a number of different periods preceding the event. In the case of Swedish and foreign government bonds, for example, we compute changes in yields over a 30 day, 60 day and 90 day window. For Swedish and foreign GDP, we look at one and two quarter growth rates. We then proceed to compute correlations with our seven different series of interest rate surprises, presented in Table C.1. Here, we simply conclude that we establish statistically significant correlations in the case of the 30, 60 and 90-day changes of EUR/SEK and USD/SEK exchange rates, as well as the 3-month changes in US and euro area inflation. We thus settle on those four variables and orthogonalize all our seven series of surprises with respect to those variables in the following way: with X_{t-} collecting the chosen variables we estimate the following regression for each of our seven monetary policy surprise series: $surprise_t = \alpha + \beta' X_{t-} + \varepsilon_t$. The resulting residual ε_t now represents the orthogonalized monetary policy surprises.

_				- 0.16						0	- 0.08						- 0.00		0.0			0.1						0.2											
0.01	-0.01	0.05	-0.07	-0.00	-0.09	0.05	0.07	0.08	0.03	0.03	-0.00	-0.20	-0.11	-0.10	-0.03	-0.05	-0.11	0.08	0.03	0.02	0.06	0.08	0.14	0.08	0.11	0.10	0.06	0.10	0.12	-0.02	0.04	0.01	-0.11	-0.08	-0.11	-0.09	0.05	0.05	10 year
0.01	-0.03	0.04	-0.08	0.02	60.0-	0.03	0.05	0.05	0.01	0.04	0.01	-0.24	-0.14	-0.13	60.0-	60.0-	-0.14	0.03	-0.02	-0.0 3	0.05	0.05	0.10	0.09	0.11	0.11	0.10	0.12	0.14	-0.01	0.06	0.02	-0.07	-0.04	-0.08	-0.06	0.04	0.06	5 year
0 00	£0.0 .	0.01	60.0-	0.06	-0.07	-0.01	0.01	0.03	-0.03	-0.02	-0.05	-0.21	-0.14	-0.15	-0.08	-0.08	-0.12	0.01	-0.05	-0.05	0.03	0.05	0.08	0.10	0.12	0.12	0.06	0.08	0.11	0.00	0.05	0.03	-0.05	-0.03	-0.06	-0.04	0.02	0.02	2 year
0.01	-0.02	0.03	-0.07	0.03	-0.05	-0.01	0.03	0.05	-0.02	-0.01	-0.04	-0.22	-0.16	-0.19	-0.13	60.0-	-0.15	0.03	-0.04	-0.05	0.04	0.07	60.0	0.10	0.13	0.15	0.10	0.11	0.15	0.03	0.08	0.07	-0.01	0.01	-0.04	00.0-	0.01	-0.00	1 year
-0 03	-0.05	0.00	-0.07	0.04	0.00	-0.05	-0.01	0.04	-0.01	-0.08	-0.05	-0.16	60.0-	-0.17	-0.08	-0.02	60.0-	0.03	-0.01	-0.03	-0.03	0.01	0.04	0.07	0.10	0.11	-0.04	-0.05	0.02	-0.03	0.04	0.05	-0.08	-0.05	-0.10	-0.05	0.01	-0.08	6 month
-0.05	-0.02	0.02	60.0-	0.18	0.01	0.02	0.06	0.11	0.07	0.05	0.06	-0.25	-0.17	-0.26	-0.22	-0.20	-0.26	0.01	-0.05	-0.07	-0.03	-0.00	0.08	0.01	0.10	0.15	0.06	0.07	0.15	-0.03	0.04	0.04	-0.01	0.02	-0.06	-0.02	0.02	-0.04	3 month
0.04	-0.03	0.03	-0.04	0.01	-0.02	-0.01	0.02	0.08	0.03	0.06	0.06	-0.23	-0.17	-0.21	-0.18	-0.19	-0.25	0.05	-0.01	-0.03	0.07	0.09	0.13	0.08	0.15	0.18	0.11	0.16	0.22	-0.06	0.02	-0.02	0.00	-0.01	-0.05	-0.04	0.00	-0.02	1 month
SE - 10V A30 -	SE - 10V A60 -	SE - 10V Δ90 -	SE - 1V A30 -	SE - 1Υ Δ60 -	SE - 1Υ Δ90 -	US - 10Y A30 -	US - 10V A60 -	US - 10Y A90 -	SE - OMXS price index A30 -	SE - OMXS price index $\Delta 60$ -	SE - OMXS price index $\Delta 90$ -	EUR/SEK A30	EUR/SEK A60	EUR/SEK A90	USD/SEK A30 -	USD/SEK 260	USD/SEK A90	Global energy prices log30	Global energy prices log60 -	Global energy prices log90	SE - inflation $\Delta 1m$ -	SE - inflation $\Delta 2m$ -	SE - inflation $\Delta 3m$ -	EU - inflation Δ1m -	EU - inflation Δ2m -	EU - inflation $\Delta 3m$ -	US - inflation $\Delta 1m$ -	US - inflation $\Delta 2m$ -	US - inflation $\Delta 3m$ -	SE - employment $\Delta 1m$ -	SE - employment Δ2m -	SE - employment $\Delta 3m$ -	SE - GDP Alg -	SE - GDP A2q -	EU - GDP A1q -	EU - GDP A2q -	US - GDP Δ1q -	US - GDP A29 -	

Table C.1 Correlations between macro variables and calculated surprises

Note: The table shows calculated pairwise correlations between economic and financial variables and monetary policy surprises based on the seven Swedish instruments (1-month OIS, 3-month OIS, 6-month OIS, 1-year swap, 2-year swap, 5-year swap and the 10-year swap). Correlations in bold are statistically significant. For variables with revision history, such as GDP and employment, we use the vintage of the data that was available one day before the monetary policy announcement. Numbers following the delta (Δ) symbol represent the number of days (30, 60 or 90), months (1m, 2m or 3m) or quarters (1q, 2q) for which the change in the variable is calculated. For example, SE – 10y Δ 30, is the percentage change in 10 year Swedish government bond yield during the 30 days that proceed a monetary policy announcement.

APPENDIX D

The table shows some summary statistics of the calculated surprises, in total and separately for positive surprises and negative surprises. In some cases, the surprise is exactly zero, hence $N_{pos} + N_{neg} < N_{tot}$.

			Positive	e surprises						
	1m	3m	6m	1y	2y	5y	10y			
Ν	105	124	122	108	106	105	109			
Mean	0.02	0.02	0.02	0.02	0.02	0.01	0.01			
Std	0.03	0.03	0.03	0.03	0.02	0.02	0.01			
Min	0.00	0.00	0.00	0.00	0.00	0.00	0.00			
Max	0.15	0.16	0.17	0.21	0.17	0.12	0.07			
	Negative surprises									
	1m	3m	6m	1y	2y	5y	10y			
Ν	90	96	111	139	144	149	146			
Mean	-0.03	-0.03	-0.02	-0.02	-0.02	-0.02	-0.01			
Std	0.07	0.05	0.04	0.03	0.03	0.02	0.01			
Min	-0.51	-0.29	-0.22	-0.19	-0.16	-0.14	-0.08			
Max	0.00	0.00	0.00	0.00	0.00	0.00	0.00			
		All surpr	ises (both 1	iegative ar	ıd positive)				
	1m	3m	6m	1y	2y	5y	10y			
Ν	265	265	265	265	265	265	265			
Mean	0.00	0.00	0.00	0.00	0.00	0.00	0.00			
Std	0.05	0.04	0.04	0.03	0.03	0.03	0.02			
Min	-0.51	-0.29	-0.22	-0.19	-0.16	-0.14	-0.08			
Max	0.15	0.16	0.17	0.21	0.17	0.12	0.07			

Table D.1 Descriptive statistics of calculated surprises (percentage points)

Note. Surprises are calculated as the difference between the median price 20 minutes exactly after the time of the monetary policy communication and the median price 10 minutes exactly before the same time.

APPENDIX E

We investigate the price movements of the krona exchange rate, an asset that can be expected to react in a predicable way to surprises in the Riksbank's actions and communication. Figure E.3a shows the reaction in our first factor, the target factor, and the SEK/USD and SEK/EUR exchange rates, respectively.



Note: The figure plots exchange rate movements in a 30-minute window around the Riksbank's monetary policy communications and the target factor. The charts on the left plot changes in SEK/USD exchange rate and the charts on the right plot SEK/EUR. The full sample contains all 265 dates (Figure 3a). Sub-sample I (Figure 3b) only includes dates of the target factor where the target factor and path factor have the same sign (i.e. move in the same direction). Sub-sample II (Figure 3c) only includes dates of the target factor and path factor have the same sign but the quantitative measures factor have the opposite sign.

Roughly half of the observations fall into the two quadrants (the first and fourth) that theory would predict, given that this factor may indeed be interpreted as a monetary policy shock (see Figure E.3a and Table E.2). In the case of the SEK/USD exchange rate, 66 per cent of the observations exhibit the expected reaction, whereby a positive (negative) shock to the target factor would be associated with an appreciation (depreciation) of the krona.

	5	1	1
	Full sample	Sub-Sample I	Sub-Sample II
SEK/USD	0.66	0.68	0.72
SEK/EUR	0.66	0.68	0.72

Table E.1. The share of observations in the expected quadrants

Note: The table presents the share of observations (dates) where we observe (i) a positive policy rate shock and an appreciation of the krona or (ii) a negative policy rate shock and a depreciation of the krona. The full sample contains all 265 dates. Sub-sample I only include dates where the target factor and path factor have the same sign (i.e. move in the same direction). Sub-sample II only include dates where the target factor and oath have the same sign but the quantitative measures factor have an opposite sign. The shares are in decimals.

However, before drawing any conclusions about the validity of the target factor as a monetary policy shock from these data visualisations, it is essential to recognize the multidimensional character of the Riksbank's communications. Given the assumptions that underpin the factor model, and the estimated factor loadings presented in Table 1, it is reasonable to interpret the financial market surprise in each of our 265 communication events as the consequence of three different shocks. For each communication event, there are three different shock realizations, corresponding to the three extracted factors. Foreign exchange market participants can be expected to react to and draw conclusions from all three of these realizations: the current level of the Riksbank policy rate (factor 1), any forward guidance, intentional or not, that is communicated concerning the future level of the policy rate (factor 2), as well as any news about current or future expected quantitative measures (factor 3).

Figure E.3b and E.3c show the reaction of the krona exchange rate in two subsamples of our 265 communication events. Figure E.3b shows how the krona exchange rate reacted in events where both factor 1 and factor 2 have realizations of the same sign, implying that the surprises to the current policy rate and to forward guidance were both either hawkish or dovish. As can be seen when comparing Figure E.3a and E.3b, a slightly higher share of observations fall into quadrants one and four in this subsample. The relevant shares are summarized in Table E.1: 68 per cent of observations fall in the two relevant quadrants. These shares increase further when looking at the subsample of events where factors 1 and 2 both have the same sign and where, in addition, factor 3 has a sign opposite to that of factors 1 and 2. Recall that a positive

shock to factor 1 and 2 increases rates whereas a positive shock to factor 3 both increases but also decreases rates at longer maturities.

We interpret this evidence on the krona exchange rate as an indication that the three extracted factors do indeed capture relevant dimensions of Riksbank communication. The monotonically increasing shares shown in Table E.1 suggest that when additional dimensions of central bank communication are accounted for, over and above communication concerning the current level of the policy rate, we can make sense of greater shares of the foreign exchange market reactions.

APPENDIX F

In this section, the data used in the macroeconomic model is described. The macroeconomic model contains six or seven endogenous variables and four exogenous variables, excluding intercept and dummies. Table F.1 reports sources and transformations of all variables that goes into the macroeconomic model. All variables are in log levels. The real interest rate is constructed by taking the log effective exchange rate plus the log CPI foreign price level, and subtracting the Swedish price level. The FCI consists of an unweighted average of three components of the FCI index described in Alsterlind et al. (2020): the housing market, the bond market and the stock market. The foreign variables are composed as a trade weighted average of the US and the euro area, so called KIX2. For the interest rate, it is composed by adding the rates in level, weighted with the KIX2 weights. For industrial production and CPI, the series are first normalizes so that the starting point takes value 100. The weighted index is then calculated using the per cent increase in the price level for each region, weighted with the KIX2-weights. The two series (CPI and industrial production) are then logged. Figure F.1 shows all the series that are used in the model estimation.

Swedish data series	Source	Transformation
2-year swap rate	Bloomberg	end-of-month value
Policy rate	Sveriges Riksbank	end-of-month value
Price level (CPIF)	Statistics Sweden	Seasonal adjustment using X-11 starting from 1995. Nat log
GDP	Statistics Sweden	Nat log
Unemployment ages 16-64	Swedish employment agency	-
Exchange rate	Sveriges Riksbank	Nat log
Financial conditions index	Sveriges Riksbank	Sum of housing, stock market and bond components
Foreign data series	Source	Transformation
Federal funds rate	Federal reserve	-
EONIA rate check	European central bank	-
Euro area industrial production	Euro stat	Nat log
US industrial production	Federal reserve	Nat log
US CPI	U.S. Bureau of Labor Statistics	Seasonal adjustment using X-11 starting from 1995. Nat log
Euro area HICP	Eurostat	Seasonal adjustment using X-11 starting from 1995. Nat log
Euro area KIX weight	Sveriges Riksbank	-
US KIX weight	Sveriges Riksbank	-
Global energy price	IMF	Nat log

Table F.1: Data series used in the estimation of the macroeconomic model



Figure F.1: Time series data used in the macroeconomic model

APPENDIX G

In this section, we report additional robustness exercises with a focus on changes in the macroeconomic model specification. First, however, we report a robustness exercise with a different cleaning methodology. In section 4, we showed that orthogonalizing the monetary policy surprises by cleaning them from ex ante correlated variables known to the public did not change the results in any economically meaningful way. It might however be the case that the correlations differ between minutes and monetary policy decisions. Therefore, we perform a second cleaning exercise in this section, where we clean the two types of announcements separately from each other.

Separate cleaning of minutes and rate announcements

Figure G.1 shows the resulting IRFs from the separate cleaning exercise. The results are qualitatively similar to the baseline case, but effects are stronger. The strength of the instrument for the target factor is however lower in this exercise, 8.5. For the second factor, F-statistics are instead higher, 10.6, implying a strong enough instrument for these shocks. Results are also qualitatively similar to the baseline case, but the effect on inflation and output are weaker.

Sample ending in 2019

Figure G.2 shows the resulting IRFs from stopping the sample in 2019 in the VAR model. The F-statistics for the target factor and the path factor are 34.1 and 8.5 respectively.

12 lags in the estimation

Figure G.3 shows the resulting IRFs from including 12 lags instead of 9 in the VAR model. The F-statistics for the target factor and the path factor are 19.5 and 5.7 respectively.

6 lags in the estimation

Figure G.4 shows the resulting IRFs from including 6 lags instead of 9 in the VAR model. The F-statistics for the target factor and the path factor are 30.6 and 6.9 respectively.

No pandemic dummies

Figure G.5 shows the resulting IRFs when removing the pandemic dummies that are included in the baseline specification on the model. The F-statistics for the target factor and the path factor are 30.4 and 9.7 respectively.

No Swedish financial conditions in estimation

In this exercise, we remove the financial conditions index from the estimation, to see how it affects the results. The results are illustrated in Figure G.6, and the F-statistics for the target factor and the path factor are 15.0 and 8.4 respectively.

CPIFxe instead of CPIF in the estimation

In the baseline estimation, we use the CPIF price level. In this robustness exercise, we exchange CPIF with CPIF excluding energy. The results are illustrated in Figure G.7, and the F-statistics for the target factor and the path factor are 23.7 and 6.1 respectively.

Figure G.1: Robustness – effect on macroeconomic variables when surprises are orthogonalized, separate cleaning for minutes and rate announcements



Panel A: Effect of target factor shock

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Figure G.2: Robustness - effect on macroeconomic variables when sample ends in 2019



Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Figure G.3: Robustness - effect on macroeconomic variables with 12 lags instead of 9



Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Figure G.4: Robustness - effect on macroeconomic variables with 6 lags instead of 9



Panel A: effect of target factor shock

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.



Figure G.5: Robustness – no pandemic dummies included in the estimation

Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.





Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Figure G.7: Robustness – CPIF excluding energy in estimation instead of CPIF



Note: The policy rate and the 2-year swap rate are measured at their end-of-period values. The dashed red lines show the point estimates in the baseline estimation. The solid lines show the point estimates of the impulse responses functions normalized to yield a 1 pp. higher policy rate on impact for the target factor shock, and 1 pp. higher 2-year swap rate on impact for the path factor shock. The shaded areas show 68 per cent bootstrapped confidence intervals around the point estimates and the dotted lines show the edges of the 90 per cent confidence intervals. The policy rate, 2-year swap rate, and unemployment rate are measured in percentage point deviations. Financial conditions are measured in index point deviations. The rest of the series are measured as per cent deviations.

Recent Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see <u>www.riksbank.se</u>

The Macroeconomic Effects of Trade Tariffs: Revisiting the Lerner Symmetry Result by Jesper Lindé and Andrea Pescatori	2019:363
Biased Forecasts to Affect Voting Decisions? The Brexit Case by Davide Cipullo and André Reslow	2019:364
The Interaction Between Fiscal and Monetary Policies: Evidence from Sweden by Sebastian Ankargren and Hovick Shahnazarian	2019:365
Designing a Simple Loss Function for Central Banks: Does a Dual Mandate Make Sense? by Davide Debortoli, Jinill Kim and Jesper Lindé	2019:366
Gains from Wage Flexibility and the Zero Lower Bound by Roberto M. Billi and Jordi Galí	2019:367
Fixed Wage Contracts and Monetary Non-Neutrality by Maria Björklund, Mikael Carlsson and Oskar Nordström Skans	2019:368
The Consequences of Uncertainty: Climate Sensitivity and Economic Sensitivity to the Climate by John Hassler, Per Krusell and Conny Olovsson	2019:369
Does Inflation Targeting Reduce the Dispersion of Price Setters' Inflation Expectations? by Charlotte Paulie	2019:370
Subsampling Sequential Monte Carlo for Static Bayesian Models by David Gunawan, Khue-Dung Dang, Matias Quiroz, Robert Kohn and Minh-Ngoc Tran	2019:371
Hamiltonian Monte Carlo with Energy Conserving Subsampling by Khue-Dung Dang, Matias Quiroz, Robert Kohn, Minh-Ngoc Tran and Mattias Villani	2019:372
Institutional Investors and Corporate Investment by Cristina Cella	2019:373
The Impact of Local Taxes and Public Services on Property Values by Anna Grodecka and Isaiah Hull	2019:374
Directed technical change as a response to natural-resource scarcity by John Hassler, Per Krusell and Conny Olovsson	2019:375
A Tale of Two Countries: Cash Demand in Canada and Sweden by Walter Engert, Ben Fung and Björn Segendorf	2019:376
Tax and spending shocks in the open economy: are the deficits twins? by Mathias Klein and Ludger Linnemann	2019:377
Mind the gap! Stylized dynamic facts and structural models by Fabio Canova and Filippo Ferroni	2019:378
Financial Buffers, Unemployment Duration and Replacement Labor Income by Mats Levander	2019:379
Inefficient Use of Competitors' Forecasts? <i>by André Reslow</i>	2019:380
How Much Information Do Monetary Policy Committees Disclose? Evidence from the FOMC's Minutes and Transcripts by Mikael Apel, Marianna Blix Grimaldi and Isaiah Hull	2019:381
Risk endogeneity at the lender/investor-of-last-resort by Diego Caballero, André Lucas, Bernd Schwaab and Xin Zhang	2019:382
Heterogeneity in Households' Expectations of Housing Prices – Evidence from Micro Data by Erik Hjalmarsson and Pär Österholm	2019:383
Big Broad Banks: How Does Cross-Selling A Affect Lending? by Yingjie Qi	2020:384
Unemployment Fluctuations and Nominal GDP Targeting by Roberto Billi	2020:385
FAQ: How do I extract the output gap? <i>by Fabio Canova</i>	2020:386

Drivers of consumer prices and exchange rates in small open economies by Vesna Corbo and Paola Di Casola	2020:387
TFP news, stock market booms and the business cycle: Revisiting the evidence with VEC models by Paola Di Casola and Spyridon Sichlimiris	2020:388
The costs of macroprudential deleveraging in a liquidity trap by Jiaqian Chen, Daria Finocchiaro, Jesper Lindé and Karl Walentin	2020:389
The Role of Money in Monetary Policy at the Lower Bound by Roberto M. Billi, Ulf Söderström and Carl E. Walsh	2020:390
MAJA: A two-region DSGE model for Sweden and its main trading partners by Vesna Corbo and Ingvar Strid	2020:391
The interaction between macroprudential and monetary policies: The cases of Norway and Sweden by Jin Cao, Valeriya Dinger, Anna Grodecka-Messi, Ragnar Juelsrud and Xin Zhang	2020:392
Withering Cash: Is Sweden ahead of the curve or just special? by Hanna Armelius, Carl Andreas Claussen and André Reslow	2020:393
Labor shortages and wage growth by Erik Frohm	2020:394
Macro Uncertainty and Unemployment Risk by Joonseok Oh and Anna Rogantini Picco	2020:395
Monetary Policy Surprises, Central Bank Information Shocks, and Economic Activity in a Small Open Economy <i>by Stefan Laséen</i>	2020:396
Econometric issues with Laubach and Williams' estimates of the natural rate of interest by Daniel Buncic	2020:397
Quantum Technology for Economists by Isaiah Hull, Or Sattath, Eleni Diamanti and Göran Wendin	2020:398
Modeling extreme events: time-varying extreme tail shape by Bernd Schwaab, Xin Zhang and André Lucas	2020:399
The Effects of Government Spending in the Eurozone by Ricardo Duque Gabriel, Mathias Klein and Ana Sofia Pessoa	2020:400
Narrative Fragmentation and the Business Cycle by Christoph Bertsch, Isaiah Hull and Xin Zhang	2021:401
The Liquidity of the Government Bond Market – What Impact Does Quantitative Easing Have? Evidence from Sweden by Marianna Blix Grimaldi, Alberto Crosta and Dong Zhang	2021:402
Five Facts about the Distributional Income Effects of Monetary Policy by Niklas Amberg, Thomas Jansson, Mathias Klein and Anna Rogantini Picco	2021:403
When domestic and foreign QE overlap: evidence from Sweden by Paola Di Casola and Pär Stockhammar	2021:404
Dynamic Macroeconomic Implications of Immigration by Conny Olovsson, Karl Walentin, and Andreas Westermark	2021:405
Revisiting the Properties of Money by Isaiah Hull and Or Sattath	2021:406
The cost of disinflation in a small open economy vis-à-vis a closed economy by Oleksandr Faryna, Magnus Jonsson and Nadiia Shapovalenko	2021:407
On the Performance of Cryptocurrency Funds by Daniele Bianchi and Mykola Babiak	2021:408
The low-carbon transition, climate commitments and firm credit risk by Sante Carbone, Margherita Giuzio, Sujit Kapadia, Johannes Sebastian Krämer, Ken Nyholm and Katia Vozian	2022:409
Seemingly Irresponsible but Welfare Improving Fiscal Policy at the Lower Bound by Roberto M. Billi and Carl E. Walsh	2022:410
Pension Reform and Wealth Inequality: Evidence from Denmark by Torben M. Andersen, Joydeep Bhattacharya, Anna Grodecka-Messi and Katja Mann	2022:411

Inflation Targeting or Fiscal Activism? by Roberto M. Billi	2022:412
Trading volume and liquidity provision in cryptocurrency markets by Daniele Bianchi, Mykola Babiak and Alexander Dickerson	2022:413
DISPERSION OVER THE BUSINESS CYCLE: PASSTHROUGH, PRODUCTIVITY, AND DEMAND by Mikael Carlsson, Alex Clymo and Knut-Eric Joslin	2022:414
Electoral Cycles in Macroeconomic Forecasts by Davide Cipullo and André Reslow	2022:415
The Curious Incidence of Monetary Policy Across the Income Distribution by Tobias Broer, John Kramer and Kurt Mitman	2022:416
Central Bank Mandates and Monetary Policy Stances: through the Lens of Federal Reserve Speeches by Christoph Bertsch, Isaiah Hull, Robin L. Lumsdaine, and Xin Zhang	2022:417
The Political Costs of Austerity by Ricardo Duque Gabriel, Mathias Klein and Ana Sofia Pessoa	2022:418
Central bank asset purchases: Insights from quantitative easing auctions of government bonds by Stefan Laséen	2023:419
Greenflation? by Conny Olovsson and David Vestin	2023:420
Effects of foreign and domestic central bank government bond purchases in a small open economy DSGE model: Evidence from Sweden before and during the coronavirus pandemic by Yildiz Akkaya, Carl-Johan Belfrage, Paola Di Casola and Ingvar Strid	2023:421
Dynamic Credit Constraints: Theory and Evidence from Credit Lines* by Niklas Amberg, Tor Jacobson, Vincenzo Quadrini and Anna Rogantini Picco	2023:422
Stablecoins: Adoption and Fragility by Christoph Bertsch	2023:423
CBDC: Lesson from a Historical Experience by Anna Grodecka-Messi and Xin Zhang	2023:424
Do Credit Lines Provide Reliable Liquidity Insurance? Evidence from Commercial-Paper Backup Lines by Niklas Amberg	2023:425
Price Pass-Through Along the Supply Chain: Evidence from PPI and CPI Microdata by Edvin Ahlander, Mikael Carlsson and Mathias Klein	2023:426
Cash for Transactions or Store-of-Value? A comparative study on Sweden and peer countries by Carl Andreas Claussen, Björn Segendorf and Franz Seitz	2023:427
Fed QE and bank lending behaviour: a heterogeneity analysis of asset purchases by Marianna Blix Grimaldi and Supriya Kapoor	2023:428
Monetary policy in Sweden after the end of Bretton Woods by Emma Bylund, Jens Iversen and Anders Vredin	2023:429
Banking Without Branches by Niklas Amberg and Bo Becker	2024:430
Climate impact assessment of retail payment services by Niklas Arvidsson, Fumi Harahap, Frauke Urban and Anissa Nurdiawati	2024:431
Four Facts about International Central Bank Communication by Christoph Bertsch, Isaiah Hull, Robin L. Lumsdaine, and Xin Zhang	2024:432
Optimal Monetary Policy with r* < 0 by Roberto Billi, Jordi Galí, and Anton Nakov	2024:433
Quantitative Easing, Bond Risk Premia and the Exchange Rate in a Small Open Economy by Jens H. E. Christensen and Xin Zhang	2024:434
Supply-Chain Finance: An Empirical Evaluation of Supplier Outcomes by Niklas Amberg, Tor Jacobson and Yingjie Qi	2024:435
Optimal Contracts and Inflation Targets Revisited by Torsten Persson and Guido Tabellini	2024:436
Potential Climate Impact of Retail CBDC Models by Niklas Arvidsson, Fumi Harahap, Frauke Urban and Anissa Nurdiawati	2024:437
Do we need firm data to understand macroeconomic dynamics? by Michele Lenza and Ettore Savoia	2024:438

2024:439
2024:440
2024:441
2024:442
2024:443
2024:444



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se